Author's response to reviews

Title: A review of injury epidemiology in the UK and Europe: some methodological considerations in constructing rates

Authors:

Roxana Alexandrescu (Roxana.Alexandrescu@postgrad.manchester.ac.uk)
Sarah J O'Brien (Sarah.O'brien@manchester.ac.uk)
Fiona E Lecky (Fiona.Lecky@manchester.ac.uk)

Version: 4 Date: 5 July 2009

Author's response to reviews: see over
MS: A review of injury epidemiology in the UK and Europe: some methodological considerations in constructing rates
MS: 5039978322581512

Dear Editors,

Thank you for your answer and for the revisions suggested regarding the literature review. Please receive the revised manuscript and the cover letter that includes responses to the editorial formatting changes requested.

Best wishes,

Roxana Alexandrescu
Formatting changes requested

Major revisions

1. The authors of this manuscript are

2. Reference 5 – for a book the publisher/city/year

3. Reference 12 - for a book the publisher/city/year

4. Table size - These must fit into an A4 portrait page.
   All the tables now fit onto a single a4 portrait page (maximum size 176mm x 230mm) and are part of the manuscript.
   The tables are complete and have been renumbered; the manuscript text has been updated to reflect the changes.
   Table 2 → Tables 2 and 3
   Table 3 → Table 4
   Table 4 → Tables 5 and 6
   Table 5 → Tables 7 and 8
   Table 6 → Tables 9 and 10
   Table 7 → Table 11

5. Reference 13 – functioning web link
   European Injury Database. https://webgate.ec.europa.eu/idbpa/welcome.jsp

6. References - References should be cited in the text using consecutive numbers. Any references not cited in the text will not be included in the final publication.
   The references are now cited in the text in consecutive numbers, all from 1 to 117. All the changes are reflected in the manuscript text and tables.

Minor revisions

7. Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals
   A review of injury epidemiology in the UK and Europe: some methodological considerations in constructing rates.
   The title has been updated in the submission system.

8. 'Email addresses' - Please remove this subheading from the title page.
   Done

9. Please delete any blank pages
   Done.

10. Tables - Please do not include vertical lines in your tables.
    Done (Table 1 updated).

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions
Done.